Author's response to reviews

Title: Improvement in health-related quality of life in osteoporosis patients treated with teriparatide

Authors:

Arthur N Lau (arthur.lau@learnlink.mcmaster.ca)
Sammy H Ali (sammy.ali@learnlink.mcmaster.ca)
Anna M Sawka (amsawka@yahoo.com)
Lehana Thabane (thabanl@mcmaster.ca)
Alexandra Papaioannou (PAPAIOANNOU@HHSC.CA)
Amiram Gafni (gafni@mcmaster.ca)
Jonathan D Adachi (jd.adachi@sympatico.ca)

Version: 4 Date: 18 September 2008

Author's response to reviews:

After reviewing the comments of Referee 3, the appropriate revisions have been made. Comments regarding analgesia use have been included in the discussion. The manuscript has also been formatted to meet the requirements of Biomed central.